Author’s response to reviews

Title: Descending Necrotizing Mediastinitis Caused by Streptococcus Constellatus in an Immunocompetent Patient: Case Report and Review of the Literature

Authors:

rui-hai ye (yehoo2003@163.com)

jun chao yang (yangjunchaozj@163.com)

hui hua hong (honghuiha@sina.com)

yu ping mao (2242723113@QQ.com)

yuan hong zhu (zhuyuan0487@126.com)

yu cao (caoyumail@163.com)

zhen wang (wangzhen610@sina.cn)

Version: 3 Date: 29 Aug 2019

Author’s response to reviews:

1. Enhanced chest computed tomography showing significantly low density for mediastinal abscesses with a reduced calcified mediastinal lymph node (arrow). Figure 2E. EBUS-TBNA of the mediastinum revealing a foam-like purulent secretion of about 5 ml (arrow). So it is clear to diagnosis DNM.

2. Yes, she had been PTB. " she had a history of tuberculosis which was regularly cured 30-years ago." The article has elaborated.

3. We adjust antibiotics according to the susceptibility of bacterial culture.

4. We have added arrows to the picture as required.

5. Relevant languages have been corrected.